Author’s response to reviews

Title: Health Care Experiences of U.S. Retirees Living in Mexico and Panama: a Qualitative Study

Authors:

Philip D Sloane (philip_sloane@med.unc.edu)
Lauren W Cohen (lauren_cohen@unc.edu)
Bryce E Haac (bryce_haac@med.unc.edu)
Sheryl Zimmerman (sheryl_zimmerman@unc.edu)

Version: 2 Date: 31 December 2012

Author's response to reviews:

In response to your request for revision, ethical approval was not obtained from Mexican and Panamanian authorities since only U.S. Citizens participated in this research. No Mexican or Panamanian nationals were involved in any way in the subject accrue or data collection.